

Geophysical Research Letters

RESEARCH LETTER

10.1029/2020GL088227

Key Points:

- We present the first global magnetosphere simulation to reveal ballooning-interchange instability of a narrow B_z minimum in the near-Earth magnetotail
- The instability is prominent during the substorm growth phase and generates earthward entropy bubbles with embedded magnetic fronts
- The bubbles drive mesoscale ionospheric field-aligned currents and auroral structures (beads) with properties matching to those observed

Supporting Information:

- Supporting Information S1
- Movie S1
- Movie S2

Correspondence to:

K. A. Sorathia, kareem.sorathia@gmail.com

Citation:

Sorathia, K. A., Merkin, V. G., Panov, E. V., Zhang, B., Lyon, J. G., Garretson, J., et al. (2020). Ballooning-interchange instability in the near-Earth plasma sheet and auroral beads: Global magnetospheric modeling at the limit of the MHD approximation. *Geophysical Research Letters*, 47, e2020GL088227. https://doi.org/10.1029/2020GL088227

Received 3 APR 2020 Accepted 12 JUN 2020 Accepted article online 15 JUN 2020

©2020. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Ballooning-Interchange Instability in the Near-Earth Plasma Sheet and Auroral Beads: Global Magnetospheric Modeling at the Limit of the MHD Approximation

K. A. Sorathia¹, V. G. Merkin¹, E. V. Panov², B. Zhang^{3,5}, J. G. Lyon⁴, J. Garretson¹, A. Y. Ukhorskiy¹, S. Ohtani¹, M. Sitnov¹, and M. Wiltberger⁵

¹The Johns Hopkins University Applied Physics Laboratory, Laurel, MD, USA, ²Space Research Institute, Austrian Academy of Sciences, Graz, Austria, ³Department of Earth Sciences, University of Hong Kong, Hong Kong, ⁴Department of Physics and Astronomy, Dartmouth College, Hanover, NH, USA, ⁵High Altitude Observatory, National Center for Atmospheric Research, Boulder, CO, USA

Abstract Explosive magnetotail activity has long been understood in the context of its auroral manifestations. While global models have been used to interpret and understand many magnetospheric processes, the temporal and spatial scales of some auroral forms have been inaccessible to global modeling creating a gulf between observational and theoretical studies of these phenomena. We present here an important step toward bridging this gulf using a newly developed global magnetosphere-ionosphere model with resolution capturing $\leq 30 \text{ km}$ azimuthal scales in the auroral zone. In a global magnetohydrodynamic (MHD) simulation of the growth phase of a synthetic substorm, we find the self-consistent formation and destabilization of localized magnetic field minima in the near-Earth magnetotail. We demonstrate that this destabilization is due to ballooning-interchange instability which drives earthward entropy bubbles with embedded magnetic fronts. Finally, we show that these bubbles create localized field-aligned current structures that manifest in the ionosphere with properties matching observed auroral beads.

Plain Language Summary The aurora has long been used as a window onto the magnetosphere. However, auroral observations are inherently limited in trying to reconstruct global magnetospheric dynamics from the "magnetic shadow" they cast on Earth. For this reason modeling has been used in tandem with observations to better contextualize and understand the data. Substorms, the violent reconfiguration of the magnetotail and one of the most dynamic magnetospheric phenomena, have been known to be preceded by the formation of bead-like structures in the aurora. The processes responsible for auroral beading and their causal versus correlative role with substorm onset have remained an enduring mystery. The vast disparity between the spatial scales of auroral beads and those of the global magnetosphere has greatly complicated the use of modeling in unraveling this mystery. We show here for the first time a demonstration of the self-consistent formation of a magnetospheric configuration that becomes unstable during the period preceding the substorm onset and that this instability manifests in the ionosphere with similar morphology to auroral beads. The global context of the model shows that the magnetospheric processes responsible for beading are not necessarily causal to onset but a consequence of the slow magnetotail reconfiguration that precedes onset.

1. Introduction

The existence of a minimum of the equatorial northward magnetic field component, B_z , in the Earth's magnetotail has been suggested theoretically as a feature of steady adiabatic convection (e.g., Hau, 1991). Such a minimum forms as plasma sheet convection proceeds and the magnetic field of the inner tail becomes progressively more stretched (e.g., Wolf et al., 2009). The presence of a B_z minimum implies a tailward B_z gradient further from the Earth, and such gradients, in turn, can be unstable to a number of plasma instabilities, including tearing (Sitnov & Schindler, 2010) or its magnetohydrodynamic (MHD) analog (Birn et al., 2018; Merkin et al., 2015; Merkin & Sitnov, 2016), flapping (Erkaev et al., 2008), or ballooning/interchange (BI) (e.g., Pritchett & Coroniti, 2011). Because of their important role in the plasma sheet dynamics, B_z

minima and tailward gradients have received significant interest recently (see Sitnov, Birn, et al., 2019, for a review).

Measuring magnetic gradients in situ is notoriously difficult, but low values of the magnetic field are indeed occasionally observed in the near-Earth tail, particularly in the transition region from a dipolar to a tail-like configuration (9–12 R_E from Earth), suggesting the presence of a B_z minimum (Ohtani & Motoba, 2017). Panov and Pritchett (2018) used a fortuitous alignment of five Time History of Events and Macroscale Interactions during Substorms (THEMIS) probes to suggest that such regions may persist for multiple hours, be azimuthally extended (up to $10 R_E$) and radially localized (within 2.5 R_E), and exhibit signatures of BI instability ("interchange heads"). Sergeev et al. (2018) inferred a B_z minimum in the near-Earth tail from observations of isotropic energetic electrons at low altitudes equatorward of an anisotropic precipitation region, suggesting a magnetic field enhancement further tailward. Large-scale tailward B_z gradients have also been detected in data mining-based reconstructions of the geomagnetic field during substorms (Stephens et al., 2019).

In this Letter we demonstrate the formation of a radially narrow ($\leq 0.5 R_{\rm E}$) B_z minimum, and the corresponding local flux tube entropy depletion, in the near-Earth magnetotail during a substorm growth phase. Such a spatially localized feature would be particularly difficult to detect using any of the above data-based techniques. Instead, we employ our newly developed global magnetosphere magnetohydrodynamic (MHD) model, Grid Agnostic MHD for Extended Research Applications (GAMERA). A distinguishing feature of the model (see section 2) is its high resolving power, enabled by sophisticated numerical algorithms (Zhang et al., 2019), which allowed us to not only reveal the B_z minimum but also demonstrate its dynamic nature due to the BI motion of magnetic flux tubes. These dynamics result in the generation of localized perturbations in ionospheric field-aligned currents and corresponding auroral signatures commonly referred to as beads (e.g., Henderson, 2009; Motoba et al., 2012; Nishimura et al., 2016). A link between the interchange heads in the plasma sheet (or entropy bubbles, since they are depleted of flux tube entropy) and localized auroral structures (beads) has been demonstrated recently using fully kinetic simulations (Panov et al., 2019). Here, we present an independent corroboration of this connection, with the difference that both the plasma sheet and the ionosphere are included as part of our global simulation, which encompasses the entirety of solar wind-magnetosphere-ionosphere interaction and thus avoids complications resulting from artificially imposed boundary conditions in regional magnetotail models.

2. Model Description and Simulation Setup

The results presented here are the first published application of our newly developed MHD model, GAMERA, to simulation of Earth's magnetosphere-ionosphere system. GAMERA was designed to be a successor to the Lyon-Fedder-Mobarry (LFM) model (Lyon, Fedder, et al., 2004), which has been at the fore-front of space physics modeling since its initial development in the early 1980s (Lyon, Brecht, et al., 1980). The GAMERA development has been guided by the goal of preserving the core numerical philosophy of LFM, characterized by high-order spatial reconstruction, aggressive flux-limiting, and intrinsically divergence-free magnetic field updates on arbitrary hexahedral grids, while modernizing and improving the algorithms and implementation. The numerical details of the core MHD algorithm, as well as the results of a suite of standard MHD test problems, have been recently presented in exhaustive detail by Zhang et al. (2019). The extension of the MHD kernel to a full magnetospheric model requires numerous additional modules, for example, electrostatic coupling to the ionosphere, ionospheric conductance models, imposition of the upstream solar wind boundary condition, and treatment of the geomagnetic background field. In this regard, our approach is broadly similar to LFM (Lyon, Fedder, et al., 2004; Merkin & Lyon, 2010), and will be described in more detail elsewhere.

To focus on the basic physical processes that occur during a substorm growth phase, we follow the long heritage of synthetic substorm simulations using the LFM model (Fedder & Lyon, 1987; Gordeev et al., 2017; Lyon et al., 1981; Slinker et al., 1995). At the same time, we leverage the resolving power of our algorithms and GAMERA's ability to efficiently use the available computational resources to simulate a global magnetosphere at a resolution approaching the ion kinetic scale. The inner boundary condition of the simulation is imposed on a spherical surface of radius $2R_{\rm E}$ centered on Earth and assumes current closure through a thin-shell ionosphere at the altitude of ~ 120 km (Merkin & Lyon, 2010). In this simulation a constant





Figure 1. A snapshot of the simulation at T=+30 min depicts the residual (with the dipole subtracted) SM B_z magnetic field in the equatorial plane. Progressive zoom-ins on the BI structure are shown. The deepest layer of zoom demonstrates the actual grid resolution and how it relates to the BI heads.

Pedersen conductance of $\Sigma_p = 10$ S was used and no Hall conductance. The magnetosphere is preconditioned using a period of alternating southward and northward interplanetary magnetic field (IMF) orientations for a period of 8 hr at which point the IMF is again turned southward ($B_z = -5$ nT) and is held fixed for the remainder of the simulation: n = 5 cm⁻³ and $V_x = -400$ km/s. We use T = 0 to refer to the time of the final southward IMF turning, that is, the beginning of the growth phase.

This simulation uses a grid with approximately 2 times the number of cells in each dimension relative to the highest resolution LFM simulation ever performed ("OCT") (Merkin, Lyon, et al., 2013; Merkin, Anderson, et al. 2013; Merkin et al., 2019; Wiltberger et al., 2015). Specifically, the grid uses 382×512×382 cells in the radial, polar, and azimuthal directions (with respect to the Solar Magnetic, SM, *X* axis). The distorted nature of the grid allows us to concentrate cells in regions of interest, for example, the plasma sheet, magnetopause and bow shock, while smoothly transitioning to coarser regions. In the central plasma sheet (near $X_{\text{SM}} \approx -20 R_{\text{E}}$) the characteristic grid lengthscale, $\ell = \sqrt[3]{\Delta V}$ where ΔV is the cell volume, is $\ell \approx 500 \text{ km}$ and $\leq 30 \text{ km}$ azimuthal resolution in the auroral ionosphere ($0.5^{\circ} \times 0.5^{\circ}$). Most critical, however, is Δz in the central plasma sheet which is 300 km, comparable to the ion kinetic scale in this region. Furthermore, the high-order spatial reconstruction allows us to capture physical features essentially at the grid scale with fewer cells than lower-order schemes would (Zhang et al., 2019). Figure 1 shows a snapshot from the simulation and, in the deepest layer of zoom, the actual equatorial grid used and how it compares to the BI heads discussed below. A similar three-dimensional visualization is shown in Figure S1 in the supporting information (SI).

3. Results

3.1. Flux Redistribution During Growth Phase

Figure 2 shows an overview of the simulation. Each panel depicts the residual (with the dipole subtracted) SM B_z magnetic field in the equatorial plane for the different times elapsed since the IMF southward turning. We first point out the high resolving power of the simulation manifested in the fine structure captured, for instance, at the magnetopause boundary due to the Kelvin-Helmholtz instability (KHI) (cf. Merkin, Lyon, et al. 2013; Sorathia et al., 2019). Figure 2a shows the simulation 15 min after the southward turning. At this time, the IMF discontinuity has already propagated down tail so it is not apparent in the figure, but the night side of the magnetosphere is still largely a reflection of its state prior to the turning. Magnetic flux has been





Figure 2. Simulation at a glance. (a–c) Snapshots of the simulation at 15 min increments after the southward IMF turning. Each panel depicts the equatorial residual, that is, deviation from dipolar, northward magnetic field and contours of B_Z =0 (green). Also shown are contours of constant magnetic latitude mapped to the equatorial magnetosphere (cyan-magenta).

accumulated during the preceding period of northward IMF and stored in the central plasma sheet and at the flanks, indicated by the positive residual fields, shades of red, in those regions.

Figure 2b depicts the state of the simulation 30 min into the growth phase. At this time, the stretching of the geomagnetic field in the near-Earth magnetotail is evident relative to the previous snapshot (Figure 2a) as well as to the beginning of the growth phase (see Figure S2 in SI). The stretching can be seen by observing the contours of constant magnetic latitude projected to the equator shown in all panels of Figure 2. It is also apparent from the more depleted magnetic field represented by the darker shades of gray color in the night-side magnetosphere. This magnetic flux depletion is a known feature of the growth phase (e.g., Sitnov, Birn, et al. 2019) which occurs due to the flux evacuation by convection to the flanks and the dayside (Gordeev et al., 2017; Hsieh & Otto, 2014; Merkin et al., 2019). Finally, Figure 2c demonstrates further depletion of the magnetic flux at the end of the growth phase. In particular, the purple contour of 72.75° magnetic latitude has moved beyond 20 $R_{\rm E}$ at midnight.

Now we draw the reader's attention to the formation of strong undulations in the magnetic field distribution on the night side due to the BI instability seen in Figure 2 and highlighted in Figure 1. These structures will be the focus of the remainder of this paper. We note here that their counterparts in the ionosphere are similar undulations in the distribution of the auroral field-aligned currents, which we return to in section 3.3. It is also worth noting that in addition to the primary, namely the inner most in Figure 1, ridge of BI "ripples" the simulation exhibits additional ridges, less pronounced and with a tendency to be more variable, up until the region is disrupted by fast flows driven by downtail reconnection. One of these ridges is clearly evident in Figure 2b, and these structures can be seen in dynamics in Movie S1 in SI. All of these ridges correspond to the tailward walls of local magnetic field minima (i.e., tailward B_z gradients) which are discussed in more detail below.

3.2. Ballooning-Interchange Instability in the Magnetosphere

Returning to Figure 1 for a moment, the inset in the upper left indicates that the azimuthal scale size of the residual B_z ripples is on the order of ~ 4,000 km, and is well resolved by the simulation. This is consistent with the observed cross-tail scales of the BI waves, corresponding to their half period, estimated to lie between 4,500 and 6,000 km (Panov et al., 2012; Panov & Pritchett, 2018). Comparison of the auroral manifestations of these magnetospheric structures with observations is presented in section 3.3.

Furthermore, Figure S2 indicates that the residual B_z ripples correspond with a localized minimum of the total equatorial magnetic field B_z , while Figures 3g–3i and Figure S3 in SI demonstrate that this (and other) B_z minima are associated with the corresponding reversed (earthward) flux tube entropy gradients. To be more precise, it is the tailward B_z gradient that corresponds to an earthward entropy gradient (c.f. Birn et al., 2018, Figures 4e,f) and thus it is these regions that become BI unstable (Bernstein et al., 1958). Here, by flux tube entropy we understand the standard expression (e.g., Birn et al., 2009), $S = \int p^{1/\gamma} dl/B$, where p is the plasma pressure, γ is the polytropic index, B is the magnetic field magnitude, and integration is performed along the flux tube from one ionospheric foot point to the other. The fact that the undulations in the equatorial B_z distribution (as well as their apparently unstable dynamics evident from Movies S1 and S2) occur in regions of inverted flux tube entropy gradients is an explicit demonstration of BI instability of these regions. Signatures of cross-tail structure expected of ballooning instability have also been reported in global simulations by Raeder et al. (2010). However, to our knowledge, the present paper is the first explicit demonstration of BI instability of localized equatorial B_z minima in a global magnetosphere simulation of a substorm growth phase according to the original energy principle criterion (Bernstein et al., 1958).

3.3. Auroral Manifestation of BI Instability

Figure 3 further concentrates on the equatorial distribution of the flux tube entropy (panels g–i) as well as the corresponding ionospheric distributions of field-aligned currents (panels d–f) and precipitating electron energy flux (panels a–c). The figure depicts the state of the simulation at three different moments of time during the growth phase, 15 min apart. The electron energy flux (Figures 3a–3c) is derived from the MHD pressure, density and field-aligned currents at the inner boundary of the simulation ($2R_E$) using the model by Fedder et al. (1995) (see also Wiltberger et al., 2009; Zhang et al., 2015), and it includes both diffuse and discrete precipitation contributions. The latter is modulated by the field-aligned currents via the Knight





Figure 3. Ionospheric consequences of BI instability in the magnetosphere. Snapshots of the simulation at 15 min increments after the southward IMF turning are shown. (a–c) Predicted precipitating electron energy flux. (d–f) Ionospheric field-aligned current density (downward positive). (g–i) Flux tube entropy with inset boxes zoomed in on BI instability heads with the former using the bottom colormap with larger dynamic range than the inset boxes. The magenta glyphs in (a) and (g) depict points in the equator and ionosphere connected by magnetic field lines. The cyan box in (c) is the region depicted more closely in Figure 4.

relation (Knight, 1973), and thus we observe significantly elevated electron fluxes in regions of upward (negative) current densities. We stress that the currents shown in Figure 3 are computed near the inner boundary of the simulation and largely retain the spatial structure of their magnetospheric driver (Figures 3g–3i). This is enabled by the high resolving power of our simulation (see Raeder et al., 2012 for discussion of magnetosphere-ionosphere mapping issues for such fine auroral structures.)

In the ionosphere, we point out the relatively narrow sheet of upward current at premidnight. This result is consistent with the simulations of the quiet growth arc using the Rice Convection Model by Yang et al. (2013). Even given the high resolution of our global simulation, the thickness of this current sheet is $\sim 1^{\circ}$ in latitude which corresponds to ~ 113 km at a 100 km altitude. This is thicker than the arc in the two-dimensional simulations by Yang et al. (2013), which afford a much higher latitudinal resolution. Still, given the morphological consistency of these upward current sheets in the two simulations, we hypothesize that we are seeing a prototypical quiet growth phase arc in our global simulation.

Figures 3g–3i demonstrate the existence of entropy bubbles at different times during the growth phase and that they are distributed in MLT, appearing in the evening, midnight, and morning sectors. Correspondingly, both R1 and R2 field-aligned currents (Figures 3d–3f) exhibit pronounced undulations throughout the night side auroral ionosphere. Movie S2 demonstrates the dynamics of this picture and additionally indicates the equatorial mapping of lines of constant latitude in the ionosphere to better match ionospheric and magnetospheric features of the simulation. The undulations in the field-aligned currents correspond to bead-like features in the discrete electron precipitation (Figures 3a–3c), which delineate localized upward currents. It is worth noting that the auroral beads in our simulation appear both in R1 and R2 current regions throughout the growth phase and thus are not necessarily onset arc features. In observations, auroral beads have also been reported both as onset arc structures (Henderson, 2009; Kalmoni et al., 2015, 2017; Nishimura et al., 2016) and not (Panov et al., 2019; Uritsky et al., 2009; Xing et al., 2020).

For a visual comparison of the simulated beads with auroral observations, Figure 4a shows Miller projection of auroral observations from the THEMIS All-Sky Imager (ASI) in Fort Yukon on 15 February 2008 at





Figure 4. (a) Miller projection of the THEMIS All-Sky Imager (ASI) at Fort Yukon (located at 66.56°N, 145.214°W) on 15 February 2008 at 09:01:51 UT. (b) A zoom-in on a region of Figure 3c (cyan box) reprocessed using a different color scale to facilitate the comparison with panel (a).

09:01:51 UT. The imager was located at 66.56°N, 145.214°W (i.e., near the middle of the bottom axis). The auroral beads were detected between about 09:01:00 and 09:02:30 UT, appeared to be about 2° wide and also separated by about 2°. The beads propagated westward at a velocity of about 0.2° per second. Figure 4b shows the results of our simulation zooming in on a region of Figure 3c spanning exactly the same distance in both longitude and latitude as Figure 4a. Even given the high resolution of our simulation, the beads in Figure 4b still appear underresolved compared to the observations, although their salient features like the size and separation are very similar to those observed. While the observations in Figure 4a are taken in the evening sector and the beads propagate westward, the beads in the simulation are more pronounced in the morning sector R2 current branch (Figures 3a–3c). This is because this auroral region both maps to the BI unstable region in the magnetosphere and has the field-aligned current of the right sense (upward) to produce discrete electron precipitation (see Movie S2 for more explicit mapping demonstration.) Still, Figures 3c, 3f, and 3i exhibit bead-like perturbations of the thin R1-sense current sheet in the evening sector, which we interpreted above as the analog of the growth phase arc discussed in the context of RCM simulations by Yang et al. (2013).

4. Discussion and Conclusions

Figure S2 demonstrates that the narrow B_z minimum appears in the simulation even prior to the beginning of the growth phase (T = 0 min). Thus, it is a persistent feature of the simulated magnetosphere, including during northward IMF conditions. It is therefore appropriate to ask why the apparent instability of this B_z dip is more pronounced during the growth phase (which is made rather evident by Movie S2). Maltsev and Mingalev (2000) have computed the BI growth rate for a minimum B_z configuration and showed that it is proportional to the equatorial plasma β_e ($\gamma_{BI} \propto \beta_e$; see their equation 22). Figure S4 shows the radial profiles of β_e extracted from the simulation at different times in the format similar to Figures S1 and S2. Figure S4 demonstrates that over the first 40 min of the growth phase, β_e grows by a factor of ~ 4–5 in the region of the B_z minimum which, taking into account the theoretical prediction by Maltsev and Mingalev (2000), suggests why the growth rate increases in the course of the growth phase. Furthermore, using numerical simulations Zhu et al. (2004) showed that for sufficiently thin current sheets, $\approx 1R_e$, γ_{BI} is completely suppressed for $\beta_e \lesssim 10$ with a sharp growth ensuing for $\beta_e > 10$ peaking at $\beta_e \approx 30$ (see their Figure 4). In our global simulation we find qualitative agreement with these results in that with a similar current sheet thickness the onset of BI growth corresponds to $\beta_e \approx 10$.

The increase in β_e in the course of the growth phase is chiefly the result of magnetotail stretching and magnetic flux depletion. An immediate corollary of this is that the state of the inner magnetotail, that is, the level of its stretching prior to the start of the growth phase, as well as the details of the growth phase driving (e.g., the magnitude of the electric field at the subsolar magnetopause), have important implications for the BI stability of the B_z dip during the growth phase and hence for the probability of observing auroral beads and their intensity. In particular, a more stretched tail prior to the start of the growth phase would imply a stronger instability and more pronounced auroral beading, other things being equal. Furthermore, given appropriate preconditioning and driving, the presence of the B_z dip on top of the progressive depletion of the magnetic field (Figure S2) may explain occasional—but rare—in situ observations of very low B_z values in this region (Ohtani & Motoba, 2017).

An implication of the above discussion is that the presence of auroral beads in the growth phase or even immediately before onset (e.g., Kalmoni et al., 2017; Nishimura et al., 2016) is not necessarily a cause of the onset, but rather an intrinsic feature of the state of the coupled inner magnetotail-auroral ionosphere system during the growth phase. In this sense, the relationship of auroral beads to onset may be not causal, but correlative. Alternatively, Oberhagemann and Mann (2019) posit, based on theoretical considerations, that anisotropic ballooning instability can be both a source of auroral beads and a trigger of the auroral substorm onset.

Returning to the B_z minimum, in our simulation of a synthetic magnetospheric substorm it forms at $R \sim 9-10$ $R_{\rm E}$ on the night side, where R is the equatorial distance from Earth. The background geomagnetic field in that region is still relatively large (~ 7 nT) even at the end of the substorm growth phase, when it is most depleted. Therefore, even when BI instability develops nonlinearly in the B_z minimum region, reconnection (otherwise allowed due to numerical resistivity) does not ensue. This is distinctly different from fully kinetic particle-in-cell (PIC) simulations (e.g., Pritchett & Coroniti, 2011, 2013) of configurations with an imposed B_z minimum, where reconnection is initiated in the wakes of earthward moving interchange heads due to magnetic flux depletion. Such reconnection would require $\beta_e \sim 500$, which was never achieved in our simulation, where the highest β_e value was ~ 100 (Figure S3). We note, however, that the equatorial B_z value in the vicinity of the B_z dip at the end of the growth phase in our simulation (Figure S2) is $1-2\sigma$ (standard deviations) below the median statistical value cited by Ohtani and Motoba (2017). Therefore, the state of the simulated magnetotail is statistically representative of the state of the stretched tail in reality.

One other significant difference between the MHD and kinetic BI modes is their azimuthal propagation direction. In the present simulation, the drift of the BI heads can be either duskward (premidnight) or dawnward (postmidnight) with typical azimuthal speeds of O(10) km/s and are caused by MHD stresses. In the PIC simulations (Pritchett & Coroniti, 2011) the direction of the azimuthal drift is defined by the direction of motion of the cross-tail current carriers in a thin current sheet (ions or electrons). Furthermore, although the scales of the observed and GAMERA-simulated BI heads in Figure 4 are similar, there are also substantial differences in their propagation speed. Indeed, the observed azimuthal drift velocities (0.2°/s) correspond to O(100) km/s phase speed in the magnetosphere, which is close to those predicted by the PIC simulations (yielding a fraction of the ion thermal velocity). This, however, may be due to the fact that the slower-drifting MHD BI heads reported here are more elusive for spacecraft observations. Despite the differences with the specific observations (Figure 4a) and PIC simulations, we note that the results of our global MHD simulation is well within the range of statistical properties of auroral beads, including their propagation direction (can be either dawnward or duskward), speed and size (e.g., Nishimura et al., 2016). One of the interesting avenues for future work would be to investigate the effects of the feedback of a realistic ionospheric conductance model and solar wind driving (c.f. Merkin et al., 2019).

An MHD simulation at the scale presented above, namely, resolving the ion kinetic scale within a fluid model, marks a qualitative transition in global magnetosphere modeling. Going beyond these scales requires departing from ideal MHD. Describing electron-scale reconnection physics may ultimately be achieved by locally embedding fully kinetic codes within global fluid models (Chen et al., 2017). At the same time, capturing larger-scale ion kinetic processes, for example, the kinetic BI effects mentioned above or thin current



sheets (e.g., Sergeev, V., Angelopoulos, V., et al. 2011), requires developing global hybrid approaches (e.g., Lin et al., 2017; Palmroth et al., 2018) or otherwise incorporating feedback of ion kinetics in fluid models (e.g., Mignone et al., 2018). However, it is imperative that any attempt to augment a fluid model with kinetic effects first ensure that global and mesoscale fluid physics are properly captured and described all the way down to the kinetic scale. Having now demonstrated that our GAMERA model satisfies this requirement, we are confident that future work concentrating on incorporation into it of kinetic effects will be put on a firm footing.

Data Availability Statement

Data used to create the figures presented are archived and available online (via https://doi.org/10.5281/ zenodo.3835863). In addition to the postprocessed data used to generate figures stored on Zenodo, raw simulation data are preserved on the Cheyenne supercomputer and can be made available upon request.

References

- Bernstein, I. B., Frieman, E. A., Kruskal, M. D., & Kulsrud, R. M. (1958). An energy principle for hydromagnetic stability problems. Proceedings of the Royal Society of London. Series A. Mathematical and Physical Sciences, 244(1236), 17–40.
- Birn, J., Hesse, M., Schindler, K., & Zaharia, S. (2009). Role of entropy in magnetotail dynamics. *Journal of Geophysical Research*, 114, A00D03. https://doi.org/10.1029/2008JA014015
- Birn, J., Merkin, V. G., Sitnov, M. I., & Otto, A. (2018). MHD stability of magnetotail configurations with a B_z hump. Journal of Geophysical Research: Space Physics, 123, 3477–3492. https://doi.org/10.1029/2018JA025290
- Chen, Y., Tóth, G., Cassak, P., Jia, X., Gombosi, T. I., Slavin, J. A., et al. (2017). Global three-dimensional simulation of Earth's dayside reconnection using a two-way coupled magnetohydrodynamics with embedded particle-in-cell model: Initial results. *Journal of Geophysical Research: Space Physics*, 122, 10,318–10,335. https://doi.org/10.1002/2017JA024186
- Erkaev, N. V., Semenov, V. S., & Biernat, H. K. (2008). Magnetic double gradient mechanism for flapping oscillations of a current sheet. Geophysical Research Letters, 35, L02111. https://doi.org/10.1029/2007GL032277
- Fedder, J. A., & Lyon, J. G. (1987). The solar wind-magnetosphere-ionosphere current-voltage relationship. *Geophysical Research Letters*, 14(8), 880–883.
- Fedder, J. A., Slinker, S. P., Lyon, J. G., & Elphinstone, R. D. (1995). Global numerical simulation of the growth phase and the expansion onset for a substorm observed by Viking. Journal of Geophysical Research, 100(A10), 19,083–19,093.
- Gordeev, E., Sergeev, V., Merkin, V., & Kuznetsova, M. (2017). On the origin of plasma sheet reconfiguration during the substorm growth phase. *Geophysical Research Letters*, 44, 8696–8702. https://doi.org/10.1002/2017GL074539
- Hau, L.-N. (1991). Effects of steady state adiabatic convection on the configuration of the near-Earth plasma sheet, 2. Journal of Geophysical Research, 96(A4), 5591–5596.
- Henderson, M. G. (2009). Observational evidence for an inside-out substorm onset scenario. *Annales Geophysicae*, 27(5), 2129–2140. https://doi.org/10.5194/angeo-27-2129-2009
- Hsieh, M.-S., & Otto, A. (2014). The influence of magnetic flux depletion on the magnetotail and auroral morphology during the substorm growth phase. *Journal of Geophysical Research: Space Physics, 119,* 3430–3443. https://doi.org/10.1002/2013JA019459
- Kalmoni, N. M. E., Rae, I. J., Murphy, K. R., Forsyth, C., Watt, C. E. J., & Owen, C. J. (2017). Statistical azimuthal structuring of the substorm onset arc: Implications for the onset mechanism. *Geophysical Research Letters*, 44, 2078–2087. https://doi.org/10.1002/ 2016GL071826
- Kalmoni, N. M. E., Rae, I. J., Watt, C. E. J., Murphy, K. R., Forsyth, C., & Owen, C. J. (2015). Statistical characterization of the growth and spatial scales of the substorm onset arc. *Journal of Geophysical Research: Space Physics*, 120, 8503–8516. https://doi.org/10.1002/ 2015JA021470
- Knight, S. (1973). Parallel electric fields. Planetary and Space Science, 21(5), 741–750.
- Lin, Y., Wing, S., Johnson, J. R., Wang, X. Y., Perez, J. D., & Cheng, L. (2017). Formation and transport of entropy structures in the magnetotail simulated with a 3-D global hybrid code. *Geophysical Research Letters*, 44, 5892–5899. https://doi.org/10.1002/2017GL073957 Lyon, J., Brecht, S. H., Fedder, J. A., & Palmadesso, P. (1980). The effects on the Earth's magnetotail from shocks in the solar wind.
 - Geophysical Research Letters, 7(9), 721-724.
- Lyon, J. G., Brecht, S. H., Huba, J. D., Fedder, J. A., & Palmadesso, P. J. (1981). Computer simulation of a geomagnetic substorm. *Physical Review Letters*, 46(15), 1038.
- Lyon, J. G., Fedder, J. A., & Mobarry, C. M. (2004). The Lyon–Fedder–Mobarry (LFM) global MHD magnetospheric simulation code. Journal of Atmospheric and Solar-Terrestrial Physics, 66(15-16), 1333–1350.
- Maltsev, Y. P., & Mingalev, O. V. (2000). Instability of the minimum B configuration. In A. Wilson (Ed.), Fifth International Conference on Substorms (Vol. 443, pp. 397), ESA Special Publication.
- Merkin, V. G., Anderson, B. J., Lyon, J. G., Korth, H., Wiltberger, M., & Motoba, T. (2013). Global evolution of Birkeland currents on 10 min timescales: MHD simulations and observations. *Journal of Geophysical Research: Space Physics*, 118, 4977–4997. https://doi.org/10.1002/ jgra.50466
- Merkin, V. G., & Lyon, J. G. (2010). Effects of the low-latitude ionospheric boundary condition on the global magnetosphere. Journal of Geophysical Research, 115, A10202. https://doi.org/10.1029/2010JA015461
- Merkin, V. G., Lyon, J. G., & Claudepierre, S. G. (2013). Kelvin-Helmholtz instability of the magnetospheric boundary in a three-dimensional global MHD simulation during northward IMF conditions. *Journal of Geophysical Research: Space Physics*, 118, 5478–5496. https://doi.org/10.1002/jgra.50520
- Merkin, V. G., Panov, E. V., Sorathia, K. A., & Ukhorskiy, A. (2019). Contribution of bursty bulk flows to the global dipolarization of the magnetotail during an isolated substorm. *Journal of Geophysical Research: Space Physics*, 124, 8647–8668. https://doi.org/10.1029/ 2019JA026872

Acknowledgments

This work was supported by NASA Grants NNX14AH77G, NNX16AG73G, 80NSSC17K0679, 80NSSC18K1381, and 80NSSC19K0241; the Van Allen Probes contract NNN06AA01C; and the NASA DRIVE Science Center for Geospace Storms (CGS) under Grant 80NSSC20K0601. S. O. acknowledges support from NASA Grant NNX16AF74G and NSF Grant 1603028. E. V. P. acknowledges the Austrian Science Fund (FWF) I 3506-N27. We would like to acknowledge the use of computational resources (doi:10.5065/ D6RX99HX) at the NCAR-Wyoming Supercomputing Center provided by the National Science Foundation and the State of Wyoming and supported by NCAR's Computational and Information Systems Laboratory.

- Merkin, V. G., & Sitnov, M. I. (2016). Stability of magnetotail equilibria with a tailward B_z gradient. Journal of Geophysical Research: Space Physics, 121, 9411–9426. https://doi.org/10.1002/2016JA023005
- Merkin, V. G., Sitnov, M. I., & Lyon, J. G. (2015). Evolution of generalized two-dimensional magnetotail equilibria in ideal and resistive MHD. Journal of Geophysical Research: Space Physics, 120, 1993–2014. https://doi.org/10.1002/2014JA020651

Mignone, A., Bodo, G., Vaidya, B., & Mattia, G. (2018). A particle module for the pluto code. I. An implementation of the MHD-PIC equations. *The Astrophysical Journal*, 859(1), 13.

Motoba, T., Hosokawa, K., Kadokura, A., & Sato, N. (2012). Magnetic conjugacy of northern and southern auroral beads. *Geophysical Research Letters*, 39, L08108. https://doi.org/10.1029/2012GL051599

- Nishimura, Y., Yang, J., Pritchett, P. L., Coroniti, F. V., Donovan, E. F., Lyons, L. R., et al. (2016). Statistical properties of substorm auroral onset beads/rays. *Journal of Geophysical Research: Space Physics*, 121, 8661–8676. https://doi.org/10.1002/2016JA022801
- Oberhagemann, L. R., & Mann, I. R. (2019). A new substorm onset mechanism: Increasingly parallel pressure anisotropic ballooning. Geophysical Research Letters, 47, e2019GL085271. https://doi.org/10.1029/2019GL085271
- Ohtani, S., & Motoba, T. (2017). Equatorial magnetic field of the near-Earth magnetotail. *Journal of Geophysical Research: Space Physics*, 122, 8462–8478. https://doi.org/10.1002/2017JA024115
- Palmroth, M., Ganse, U., Pfau-Kempf, Y., Battarbee, M., Turc, L., Brito, T., et al. (2018). Vlasov methods in space physics and astrophysics. Living Reviews in Computational Astrophysics, 4(1), 1.
- Panov, E. V., Baumjohann, W., Nakamura, R., Pritchett, P. L., Weygand, J. M., & Kubyshkina, M. V. (2019). Ionospheric footprints of detached magnetotail interchange heads. *Geophysical Research Letters*, 46, 7237–7247. https://doi.org/10.1029/2019GL083070

Panov, E. V., & Pritchett, P. L. (2018). Dawnward drifting interchange heads in the Earth's magnetotail. *Geophysical Research Letters*, 45, 8834–8843. https://doi.org/10.1029/2018GL078482

Panov, E. V., Sergeev, V. A., Pritchett, P. L., Coroniti, F. V., Nakamura, R., Baumjohann, W., et al. (2012). Observations of kinetic ballooning/interchange instability signatures in the magnetotail. *Geophysical Research Letters*, 39, L08110. https://doi.org/10.1029/ 2012GL051668

- Pritchett, P. L., & Coroniti, F. V. (2011). Plasma sheet disruption by interchange-generated flow intrusions. *Geophysical Research Letters*, 38, L10102. https://doi.org/10.1029/2011GL047527
- Pritchett, P. L., & Coroniti, F. V. (2013). Structure and consequences of the kinetic ballooning/interchange instability in the magnetotail. Journal of Geophysical Research: Space Physics, 118, 146–159. https://doi.org/10.1029/2012JA018143
- Raeder, J., Zhu, P., Ge, Y., & Siscoe, G. (2010). Open geospace general circulation model simulation of a substorm: Axial tail instability and ballooning mode preceding substorm onset. *Journal of Geophysical Research*, 115, A00116. https://doi.org/10.1029/2010JA015876
- Raeder, J., Zhu, P., Ge, Y., & Siscoe, G. (2012). Auroral signatures of ballooning mode near substorm onset: Open geospace general circulation model simulations. Auroral Phenomenology and Magnetospheric Processes: Earth And Other Planets, 197, 389–395. https://doi.org/10.1029/2011GM001200
- Sergeev, V., Angelopoulos, V., Kubyshkina, M., Donovan, E., Zhou, X.-Z., Runov, A., et al. (2011). Substorm growth and expansion onset as observed with ideal ground-spacecraft THEMIS coverage. *Journal of Geophysical Research*, 116, A00I26. https://doi.org/10.1029/ 2010JA015689
- Sergeev, V. A., Gordeev, E. I., Merkin, V. G., & Sitnov, M. I. (2018). Does a local B-minimum appear in the tail current sheet during a substorm growth phase? *Geophysical Research Letters*, 45, 2566–2573. https://doi.org/10.1002/2018GL077183
- Sitnov, M., Birn, J., Ferdousi, B., Gordeev, E., Khotyaintsev, Y., Merkin, V., et al. (2019). Explosive magnetotail activity. Space science reviews, 215(4), 31.
- Sitnov, M. I., & Schindler, K. (2010). Tearing stability of a multiscale magnetotail current sheet. *Geophysical Research Letters*, 37, L08102. https://doi.org/10.1029/2010GL042961
- Sitnov, M. I., Stephens, G. K., Tsyganenko, N. A., Miyashita, Y., Merkin, V. G., Motoba, T., et al. (2019). Signatures of non-ideal plasma evolution during substorms obtained by mining multi-mission magnetometer data. *Journal of Geophysical Research: Space Physics*, 124, 8427–8456. https://doi.org/10.1029/2019JA027037
- Slinker, S. P., Fedder, J. A., & Lyon, J. G. (1995). Plasmoid formation and evolution in a numerical simulation of a substorm. *Geophysical Research Letters*, 22(7), 859–862. https://doi.org/10.1029/95GL00300

Sorathia, K. A., Merkin, V. G., Ukhorskiy, A. Y., Allen, R. C., Nykyri, K., & Wing, S. (2019). Solar wind ion entry into the magnetosphere during northward IMF. Journal of Geophysical Research: Space Physics, 124, 5461–5481. https://doi.org/10.1029/2019JA026728

- Stephens, G. K., Sitnov, M. I., Korth, H., Tsyganenko, N. A., Ohtani, S., Gkioulidou, M., & Ukhorskiy, A. Y. (2019). Global empirical picture of magnetospheric substorms inferred from multimission magnetometer data. *Journal of Geophysical Research: Space Physics*, 124, 1085–1110. https://doi.org/10.1029/2018JA025843
- Uritsky, V. M., Liang, J., Donovan, E., Spanswick, E., Knudsen, D., Liu, W., et al. (2009). Longitudinally propagating arc wave in the pre-onset optical aurora. *Geophysical Research Letters*, 36, L21103. https://doi.org/10.1029/2009GL040777
- Wiltberger, M., Merkin, V., Lyon, J. G., & Ohtani, S. (2015). High-resolution global magnetohydrodynamic simulation of bursty bulk flows. Journal of Geophysical Research: Space Physics, 120, 4555–4566. https://doi.org/10.1002/2015JA021080
- Wiltberger, M., Weigel, R. S., Lotko, W., & Fedder, J. A. (2009). Modeling seasonal variations of auroral particle precipitation in a global-scale magnetosphere-ionosphere simulation. *Journal of Geophysical Research: Space Physics*, 114, A01204. https://doi.org/ 10.1029/2008JA013108
- Wolf, R. A., Wan, Y., Xing, X., Zhang, J. C., & Sazykin, S. (2009). Entropy and plasma sheet transport. *Journal of Geophysical Research*, 114, A00D05. https://doi.org/10.1029/2009JA014044
- Xing, X., Wang, C.-P., Liang, J., & Yang, B. (2020). Ballooning instability in the plasma sheet transition region in conjunction with non-substorm auroral wave structures. *Journal of Geophysical Research: Space Physics*, 125, e2019JA027340. https://doi.org/10.1029/ 2019JA027340
- Yang, J., Wolf, R. A., Toffoletto, F. R., & Sazykin, S. (2013). RCM-E simulation of substorm growth phase arc associated with large-scale adiabatic convection. *Geophysical Research Letters*, 40, 6017–6022. https://doi.org/10.1002/2013GL058253
- Zhang, B., Lotko, W., Brambles, O., Wiltberger, M., & Lyon, J. (2015). Electron precipitation models in global magnetosphere simulations. Journal of Geophysical Research: Space Physics, 120, 1035–1056. https://doi.org/10.1002/2014JA020615
- Zhang, B., Sorathia, K. A., Lyon, J. G., Merkin, V. G., Garretson, J. S., & Wiltberger, M. (2019). Gamera: A three-dimensional finite-volume MHD solver for non-orthogonal curvilinear geometries. *The Astrophysical Journal Supplement Series*, 244(1), 20.
- Zhu, P., Bhattacharjee, A., & Ma, Z. W. (2004). Finite k_y ballooning instability in the near-Earth magnetotail. *Journal of Geophysical Research*, 109, A11211. https://doi.org/10.1029/2004JA010505